Author’s response to reviews

Title: Inhibition of autophagy enhances synergistic effects of Salidroside and anti-tumor agents against colorectal cancer

Authors:

Hai Li (lihaiges@hust.edu.cn)

Chen Chen (chenchenbts@yahoo.com)

Version: 1 Date: 19 Oct 2017

Author’s response to reviews:

Response to reviewer 1.

The language has been revised by a native English-speaking people, and the references have been changed to BMC Complementary and alternative medicine journal style.

Response to reviewer 2.

We rewrited the comments to the authors

Response to reviewer 3.

We added Figure 1 F in figure legend. Furthermore, we make a schematic representation of the the conclusion.